Author’s response to reviews

Title: The Improving Care in Chronic Obstructive Lung disease study: CAROL
Improving processes of care and quality of life of COPD patients in primary care:
A cluster randomized trial

Authors:

Claudia A. Steurer-Stey (claudia.stey@usz.ch)
Stefan Markun (stefan.markun@usz.ch)
Kaba Dallalana (kaba.dallalana@usz.ch)
Anja Frei (frei.anja@usz.ch)
Ulrike Held (ulrike.held@usz.ch)
Michel Wensing (m.wensing@iq.umcn.nl)
Thomas Rosemann (Thomas.rosemann@usz.ch)

Version: 5
Date: 7 March 2014

Author’s response to reviews:

Dear Editors

Please find our revised manuscript protocol

The Improving Care in Chronic Obstructive Lung disease study: CAROL
Improving processes of care and quality of life of COPD patients in primary care:
A cluster randomized trial

Michel Wensing was unintentionally listed as MH in the Authors’ Contributions
section of the manuscript. It’s corrected to MW.

We would be pleased if you considered our revised protocol for publication.

Kind regards

Claudia Steurer-Stey, MD